Author's response to reviews

Title: Mechanisms of pulmonary dysfunction after on-pump and off-pump cardiac surgery: a prospective cohort study

Authors:

A.B. Johan Groeneveld (johan.groeneveld@vumc.nl)
Evert K. Jansen (ek.jansen@vumc.nl)
Joanne Verheij (j.verheij@vumc.nl)

Version: 3 Date: 14 February 2007

Author's response to reviews:

Dear Mr. Zamvar,

It was a great pleasure to receive your mail of February 12, explaining, that after some minor adjustments, our manuscript "Mechanisms of pulmonary dysfunction after on-pump and off-pump cardiac surgery: a prospective cohort study" would be accepted for publication in your journal.

All adjustments requested by the editorial team were made:
- the text 'research article' was removed from the title page;
- the full stop was removed from the end of the title;
- each affiliation is now written in full;
- keywords section has been removed;
- references now contain all authors.

Furthermore one or two few spelling errors were found and corrected.

We hope that the manuscript now meets your standards and wish to thank you for the opportunity to publish our work in your journal,

Yours sincerely,

Johan Groeneveld